Author’s response to reviews

Title: Morphologic analysis of the proximal tibia after open wedge high tibial osteotomy for proper plate fitting

Authors:

Oui Sik Yoo (osyoo@corentec.com)
Yong Seuk Lee (smcos1@hanmail.net)
Myung Chul Lee (smcos1@naver.com)
Jae Hong Park (ParkJH@hanmail.net)
Jae Won Kim (JKKIM@corentec.com)
Doo Hoon Sun (smcos1@snu.ac.kr)

Version: 2 Date: 06 Sep 2016

Author’s response to reviews:

This is the revision of “BMSD-D-16-00455R1 Morphologic analysis of the proximal tibia after open wedge high tibial osteotomy for proper plate fitting, Oui Sik Yoo, MS; Yong Seuk Lee; Myung Chul Lee, MD, PhD; Jae Hong Park, MD; Jae Won Kim, MS; Doo Hoon Sun, BMC Musculoskeletal Disorders”

Thank you for your kind and detailed edition. I would like thank you again very greatly for your comments and contributions on making this a better article. We have changed our article as you suggested. We have highlighted the changes in red and have added a detailed description on the areas that you have suggested.

Dear Professor Lee,

Please also ensure that your revised manuscript conforms to the journal style, which can be found at the Instructions for Authors on the journal homepage.

Yes, we rechecked again and confirm that it conforms to the journal style.
By resubmitting your manuscript you confirm that all author details on the revised version are correct, that all authors have agreed to authorship and order of authorship for this manuscript and that all authors have the appropriate permissions and rights to the reported data.

Yes, it is right. We rechecked at the title page. Additionally, we described what you commented in line 325-328.

Reviewer reports:

The paper has been assessed and requires some substantial changes.

For the paper changes we recommend following the reviewers’ suggestions.

No comment was given in the reviewers’ report.

In accordance with BioMed Central editorial policies and formatting guidelines, all submissions to BMC Musculoskeletal Disorders must have a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal’s Submission Guidelines web page for information regarding the criteria for each sub-section (http://www.biomedcentral.com/bmcmusculoskeletdisord/).

Where a mandatory section is not relevant to your study design or article type, for example, if your manuscript does not contain any individual persons data, please write "Not applicable" in these sections.

We answered all about your concerns.

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must state that data will not be shared, and provide reasons for this in the manuscript text. For further guidance on how to format this section, please refer to BioMed Central's editorial policies page - http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials.

New line 318-320, we edited as you commented.

Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Authors' Information

New line 315-337, we edited as you commented.

Best regards

Sincerely yours